# The causes and challenges of low interest rates: insights from basic principles and recent literature

Youchang Wu

Department of Finance, Lundquist College of Business, University of Oregon, Eugene, Oregon, USA

# Abstract

Causes and challenges of low interest rates

Received 15 June 2020 Revised 6 August 2020 19 August 2020 Accepted 3 September 2020

**Purpose** – What causes the downward trend of real interest rates in major developed economies since the 1980s? What are the challenges of the near-zero interest and inflation rates for monetary policy? What can the policymakers learn from the latest developments in the monetary and interest rate theory? This paper aims to answer these questions by reviewing both basic principles of interest rate determination and recent academic and policy debates.

**Design/methodology/approach** – The paper critically reviews the explanations for the downward trend of real interest rates in recent decades and monetary policy options in a near-zero interest rate environment. **Findings** – The decline of real interest rates is likely an outcome of multiple technological, social and economic factors including diminished productivity growth, changing demographics, elevated tail-risk concerns, time-varying convenience yields of safe assets, increased global demand for safe assets, rising wealth and income inequality, falling relative price of capital, accommodative monetary policies, and changes in industry structure that alter the investment and saving behaviors of the corporate sector. The near-zero interest rate limits the space of central banks' response to economic crises. It also challenges some conventional wisdoms of monetary

space of central banks' response to economic crises. It also challenges theory and sparks radically new ideas about monetary policy.

**Originality/value** – This survey differs from the existing work by taking a broader view of both economics and finance literature. It critically assesses the economic forces driving the global decline of real interest rates through the lens of basic principles and empirical evidence and discusses the merits and limitations of each proposed explanation. The study emphasizes the importance of a better understanding of economic forces driving diverging trends of corporate investment and saving behaviors. It also discusses the implications of the neo-Fisherism and the fiscal theory of price level for monetary policy in a low interest rate environment.

**Keywords** Interest rate, Safe asset, Tail risk, Secular stagnation, Monetary policy, Zero lower bound **Paper type** Literature review

# 1. Introduction

There is a significant decline in real interest rates, defined as nominal rates minus expected inflation rates, in major developed economies since the 1980s. Panel (a) of Figure 1 plots the real and nominal 1-year treasury rates in the US from January 1980 to April 2020 and the real 10-year rates measured by the yield on the US Treasury Inflation-Protected Securities (TIPSs) since 2003. The downward trend is pronounced. The nominal 1-year rate declines from the peak of 16.7% per annum in August 1981 to 0.18% in April 2020, while the real 1-year rate drops from 6.9% to -2.2%. From August 2007 (the month marking the start of a series of rate cuts by the Federal Reserve System in response to the worsening subprime mortgage crisis) to the sample end, the average 1-year rate is 0.89% in nominal terms and -1.43% in real terms, while the average real 10-year rate is 0.66% per annum [1].

The secular decline of real interest rates is a global phenomenon. In fact, interest rates have been lower in many other developed countries than in the US since the 2008 financial crisis. In Japan, Switzerland and many European Union countries, even nominal interest rates have been in the negative territory in recent years. In particular, the current German



China Finance Review International © Emerald Publishing Limited 2044-1398 DOI 10.1108/CFRI-06-2020-0071

The author thanks an anonymous referee and Wenfeng Wu (Editor) for valuable comments and suggestions.





Nominal LIBOR rates for EUR, GBP, JPY, and CHF (b)

**Note(s):** Panel (a) shows the nominal and real 1-year constant maturity Treasury rates in the US from January 1980 to April 2020 and the 10-year TIPS yield from January 2003 to April 2020. The 1-year real rate is calculated as the nominal rate minus the median 12-month inflation forecast made by economists responding to the Livingston Survey. The shaded areas indicate the recession periods in the US dated by the National Bureau of Economic Research. Panel (b) shows the time series of nominal 3-month LIBOR rates for the Euro, British pound, Japanese yuan and Swiss franc up to April 2020.

**Source(s):** The Livingston Survey data, downloaded from the website of the Federal Reserve Bank of Philadelphia; all other data, retrieved from FRED, Federal Reserve Bank of St. Louis.

Figure 1. Interest rates trends government bond yields are negative for all maturities up to 30 years. Negative nominal interest rates are not only observed in government bonds. Panel (b) of Figure 1 shows the nominal 3-month London Interbank Lending Rates (LIBOR) for the euro, British pound, Japanese yuan and Swiss franc. Among them only the rate for the British pound remains above zero at the sample end. Yi and Zhang (2017) examine real interest rates in the 20 largest economies from (up to) 1955 to 2014 and document a steady decline and convergence across countries since late 1980s.

This pronounced downward trend of real interest rates in over three decades and the persistent low interest rates in recent years have posed many theoretical and practical questions. What causes the downward trend? What are the challenges of the near-zero interest and inflation rates for monetary policy? And what can the policymakers learn from the latest developments in the monetary and interest rate theory? This paper aims to answer these questions by reviewing both basic principles of interest rate determination and recent academic and policy debates. The canonical models of the neoclassical asset pricing and growth theory point to productivity growth, risk, and population growth as the main determinants of real interest rates, in addition to the subjective discount factor. Recent studies have examined the role of these factors in depth as well as factors beyond these models, including convenience yields of safe assets, global demand for safe assets, relative price of capital, wealth and income inequality, and monetary policy. I review the explanations based on these factors, I emphasize the importance of a better understanding of economic forces leading to a downward trend in corporate investment and an upward trend in corporate saving.

The near-zero nominal interest rate limits the space of central banks' response to economic crises. A series of unconventional monetary policies have been experimented following the global financial crisis in 2008, including quantitative easing, forward interest rate guidance, and negative interest rates. More aggressive actions have been taken upon the outbreak of the COVID-19 pandemic. While the monetary policies post the 2008 crisis have been successful in maintaining the stability of the financial system, they do not appear to be very effective in boosting aggregate demand and growth. In fact, most central banks consistently undershoot their inflation targets. These experiences and challenges have spurred or generated renewed interest in unconventional ideas of monetary theory, including the new Fisherism and the fiscal theory of price level. The last part of this review discusses the monetary policy conundrum in the near-zero interest rate environment and the policy implications of these new developments in monetary theory.

The literature on the low real interest rate is vast, and it is growing at a fast pace. Therefore, this is by no means a complete survey. Several papers and commentaries provide an overview of the topic. Summers (2014) discusses a list of potential contributing factors to the decline. He revives the term "secular stagnation" first introduced by Alvin Hansen and uses it as an overarching concept to describe the current prolong situation of low demand, low growth and low interest rates. Hall (2017a) discusses the causes and consequences of low interest rates. The US Council of Economic Advisors (2015) reviews the world-wide decline of long-term interest rates and discusses potential reasons. Rachel and Smith (2017) examine quantitatively the contributions of various factors to the shifting of global investment and saving schedules. This survey differs from the existing work by taking a broader view of both economics and finance literature and by critically reviewing of each proposed explanation and policy recommendation through the lens of basic principles and empirical evidence.

The review is structured as follows. Section 2 reviews the standard neoclassical theory of real interest rate determination. Section 3 reviews the recent findings on causes of the secular decline. Section 4 discusses the challenges of low interest rates for monetary policy and the policy implications of the neo-Fisherism and the fiscal theory of price level. Section 5 concludes with a short summary.

# CFRI

# 2. The determination of real risk-free rates: basic principles

In this section, I review the determination of real risk-free rates in the standard consumptionbased asset pricing theory and neoclassical growth theory. Following the tradition of the "classical dichotomy," which maintains that real variables such as the real interest rate are determined only by real factors and not by monetary factors, these standard theories abstract from potential influences of monetary policy. While the canonical models reviewed here leave out many important drivers of real interest rates, they highlight the key mechanisms and provide a foundation for models with more realistic features. The closed-form results they offer provide transparency and clarity often unavailable from the more sophisticated models in modern literature.

## 2.1 Risk-free rates in the consumption-based asset pricing theory

Consider a representative agent with an infinite horizon, whose preferences are described by a time-separable expected utility function over the stream of consumption { $C_0, C_1, C_2, \ldots$  }:

$$E_0[U(C_0, C_1, C_2, \ldots)] = E_0 \sum_{t=0}^{\infty} \beta^t u(C_t),$$

where  $\beta < 1$  is the subjective discount factor capturing the agent's patience (a higher  $\beta$  means more patience),  $u(C_t)$  is the per-period utility function with  $u'(C_t) > 0$ ,  $u''(C_t) \le 0$ . The Euler equation, which characterizes the intertemporal optimization of the representative agent, dictates that the period-*t* price of any investment asset with the payoff  $x_{t+1}$  at t + 1 should be

$$p_t = E_t \left[ \frac{\beta u'(C_{t+1})}{u'(C_t)} x_{t+1} \right]$$
(1)

The marginal utility ratio  $M_{t+1} \equiv \frac{\beta u'(C_{t+1})}{u'(C_t)}$  measures how much the agent values the consumption in period t+1 relative to the consumption in period t. It is called the stochastic discount factor. Eqn (1) is a fundamental consumption-based asset pricing formula – it is the first equation that appears in Cochrane's (2005) influential *Asset Pricing* textbook. The intuition for the formula is simple. The representative agent must decide how much to consume in the current period and how much to invest. If she purchases one extra unit of asset at the price  $p_t$ , she gives up  $p_t$  units of period-t consumption, so the marginal utility cost of the purchase is  $p_t u'(C_t)$ . In exchange, the expected marginal utility benefit from additional consumption in period-t+1 is  $E_t[u'(C_{t+1})x_{t+1}]$ . The first-order condition of optimality requires that marginal cost be equal to marginal benefit, hence Eqn (1).

Denoting the gross asset return  $x_{t+1}/p_t$  by  $R_{t+1}$ . Eqn (1) can be written as

$$E_t(M_{t+1}R_{t+1}) = 1 \tag{2}$$

The gross risk-free rate from period t to t + 1,  $R_{f,t}$ , is known in period t, so it is given by

$$R_{f,t} = \frac{1}{E_t[M_{t+1}]}.$$
(3)

Assume that the preferences can be described by the power utility function with a constant relative risk aversion coefficient  $\gamma > 0$ :  $u(C_t) = \frac{C_t^{1-\gamma} - 1}{1-\gamma}$ . Eqn (3) then implies

$$R_{f,t} = \frac{1}{\beta E_t[(C_{t+1}/C_t)^{-\gamma}]}$$
(4)

Assume that the consumption follows a log-normal process, which implies that the continuously compounded consumption growth rate,  $\Delta c_{t+1} \equiv \ln(C_{t+1}) - \ln(C_t)$ , is normally distributed. The continuously compounded risk-free rate,  $\ln(R_{f,t})$ , is then given by

$$\ln(R_{f,t}) = -\ln(\beta) + \gamma E_t(\Delta c_{t+1}) - \frac{\gamma^2}{2} \sigma_t^2(\Delta c_{t+1}),$$
(5)

where  $\sigma_t^2(\Delta c_{t+1})$  is the conditional variance of  $\Delta c_{t+1}$  [2]. Hence, the real risk-free rate is low when agents are patient ( $\beta$  close to 1), when the expected consumption growth rate is low, or when the risk is high; and the sensitivities of the risk-free rate to the expected consumption growth rate and the risk are governed by the relative risk aversion coefficient  $\gamma$ . When people are impatient or when the expected consumption growth rate is high, the marginal utility of future consumption is low, so the interest rate should increase to induce saving. By contrast, when future consumption is risky, the precautionary saving motive is strong, which lowers the equilibrium risk-free rate. Not surprisingly, the precautionary saving component of the risk-free rate is bigger in magnitude when the risk is higher and when the representative agent is more risk averse [3].

An important message of Eqn (5) is that a negative real interest rate can arise naturally when the expected consumption growth is low and when the risk is high, which is helpful for understanding the negative real interest rates recently observed in many countries.

Assume further that the gross return of a risky asset,  $R_{t+1}$ , and consumption are jointly lognormally distributed. It follows from Eqns (2) and (5) that the risk premium for this asset should be

$$E_t[\ln(R_{t+1})] + \frac{1}{2}\sigma_t^2[\ln(R_{t+1})] - \ln(R_{f,t}) = \gamma \text{COV}_t(\Delta c_{t+1}, \ln(R_{t+1})),$$
(6)

where  $\sigma_t^2[\ln(R_{t+1})]$  is the conditional variance of  $\ln(R_{t+1})$ , which accounts for the difference between the expected simple return and the expected continuously compounded return; and  $\text{COV}_t(\Delta c_{t+1}, \ln(R_{t+1}))$  is the conditional covariance. If the return is positively correlated with the consumption growth, then the asset is less attractive because it generates high returns when the marginal utility of consumption is low. Such assets should offer a risk premium in order to attract investors.

Rubinstein (1976); Lucas (1978) and Breeden (1979) pioneer the development of the consumption-based asset pricing theory. In particular, Lucas (1978) derives asset prices in an endowment economy, in which the consumption process is pinned down by an exogenous endowment process. Mehra and Prescott (1985) test a variation of the Lucas (1978) model using the US *data*. Due to the low covariance between the aggregate consumption and the stock market returns, they show that the model needs an unrealistically high degree of risk aversion, which is inconsistent with the low and stable risk-free rate historically observed, to explain the high equity premium observed in the data. This empirical challenge, known as the equity premium puzzle, has generated a vast volume of literature (see Mehra (2003) for a brief survey) [4]. One strand of the literature, represented by Rietz (1988) and Barro (2006), argues that the high equity premium and low risk-free rate can be reconciled by accounting for rare but severe disasters such as world wars, pandemics, economic crises or natural disasters in an otherwise standard representative agent model. As we will see, this insight turns out to be quite helpful for understanding the low interest rate after the Great Recession.

# 2.2 Real risk-free rates in a production economy

The consumption-based asset pricing theory describes the equilibrium relation between asset returns and consumption growth. In an endowment economy such as the one in Lucas (1978), this is sufficient to pin down the risk-free rate. However, in a more general economy with

production, the risk-free rate and the consumption process must be determined jointly in a general equilibrium. This brings us to the realm of the neoclassical growth theory, which has a long history and plays a fundamental role in dynamic macroeconomics. Two workhorse models of the neoclassical growth theory are the Ramsey–Cass–Koopman model developed by Ramsey (1928), Cass (1965), and Koopman (1965) and the overlapping generations model pioneered by Samuelson (1958) and Diamond (1965).

The Ramsey–Cass–Koopman model considers a representative household with an infinite horizon, which is endowed with an initial capital stock  $K_0$  and has access to the production function:

$$Y_t = f(K_t, A_t),$$

where  $\{A_t\}_{t\geq 0}$  is a process representing the evolution of technology (productivity). Assume a constant capital depreciation rate  $\delta$  and the absence of capital adjustment cost. The law of motion for capital is

$$K_{t+1} = Y_t - C_t + (1 - \delta)K_t.$$

The marginal return on period-*t* investment,  $R_{t+1}^K$ , is the marginal product of capital in period t + 1 plus undepreciated capital:

$$R_{t+1}^{K} = f_{K}(K_{t+1}, A_{t+1}) + (1 - \delta) \equiv \text{MPK}_{t+1} + (1 - \delta).$$
(7)

This is a special case of the investment return defined by Cochrane (1991). The optimality of investment requires the Euler Eqn (2) to hold for  $R_{t+1}^k$ :

$$E_t(M_{t+1}R_{t+1}^K) = 1.$$

In case the technology process  $\{A_t\}_{t>0}$  is deterministic, this implies

$$r_{f,t} + \delta = \mathrm{MPK}_{t+1},\tag{8}$$

where  $r_{f,t} \equiv R_{f,t} - 1$  represents the net risk-free rate. Eqn (8) shows that in the absence of uncertainty, marginal product of capital should be equal to the user cost of capital  $r_{f,t} + \delta$ .

When the technology process is random, investment in physical capital also commands a risk premium. Specifically, if  $R_{t+1}^{K}$  and  $M_{t+1}$  are jointly lognormally distributed, an assumption that is more likely to hold when  $\delta = 1$ , and if the representative agent has power utility, then the risk premium for investment in physical assets is simply given by Eqn (6), with  $R_{t+1}$  replaced by  $R_{t+1}^{K}$ .

Under the assumptions of power utility, Cobb–Douglas production function and laboraugmenting productivity growth (i.e. productivity growth has the same effect on output as labor growth does), the real interest rate in the deterministic steady state of the Ramsey– Cass–Koopman model is given by

$$\ln(R_f) = -\ln(\beta) + \gamma \ln(1+g), \tag{9}$$

where g is the simple rate of productivity growth (thus  $\ln(1+g)$  is the continuouslycompounded rate). Note that Eqn (9) is a special case of Eqn (5), because in the deterministic steady state (a state that the economy converges to in the absence of any shocks), per capita consumption and productivity grow at the same rate, and the variance term in Eqn (5) vanishes. The only difference is that Eqn (9) links the risk-free rate to an exogenous productivity growth rate, while Eqn (5) links it to an endogenous consumption growth rate.

When the productivity growth is random, the model converges to a stochastic steady state, in which consumption growth hovers around its long-run mean endogenously in response to random productivity shocks. As a result, the risk-free rate is driven by both the mean and the volatility of productivity shocks. The baseline Ramsey–Cass–Koopman model can be extended to account for population growth by allowing the size of the representative household to vary over time. It turns out that this has no effect on the real interest rate in the steady state. This is because the representative household in this model internalizes the welfare of its future generations and gives more weights to future utilities if the population growth rate is higher. This effectively increases the subjective discount factor. The resulting increase in saving neutralizes the positive effect of population growth on the real interest rate. By contrast, such altruism is absent in an overlapping generations model. As a result, the real interest rate is usually positively related to the population growth rate in an economy featuring overlapping generations.

The overlapping generations models deviate from the paradigm of a representative agent economy. In the baseline model of Diamond (1965), the economy is populated by two coexisting generations in each period – the young and the old. While the economy goes on forever, each agent lives only for two periods, working when she is young and living on savings when she is old. Under the simplifying assumptions of log utility, which is a special case of the power utility (with  $\gamma = 1$ ), Cobb–Douglas production function and laboraugmenting productivity growth, the real interest rate in the deterministic steady state of the competitive equilibrium of the Diamond (1965) model is

$$r_f = \frac{\alpha}{1-\alpha} \frac{1+\beta}{\beta} (1+g)(1+n) - \delta \tag{10}$$

where  $\alpha$  is the capital income share of output, *g* and *n* are the productivity and population growth rates, respectively [5]. Because each generation maximizes its own utility, the effective subjective discount factor does not change with the population growth rate. As a result, if the population grows at a higher rate, the steady-state capital–labor ratio decreases, raising the marginal product of capital and the interest rate. This is also the case in the Solow (1956) growth model, in which saving rate is specified exogenously.

To summarize, the standard consumption-based asset pricing theory and neoclassical growth theory identify several major drivers of real interest rates: (1) the subjective discount factor, which can vary across countries and cultures; (2) the per capita consumption growth rate, which in steady state is driven by productivity growth; (3) risk premium, which is a function of the representative agent's risk aversion and the riskiness of the productivity (or consumption) growth. Furthermore, the overlapping generations model of the neoclassical growth theory predicts a positive effect of population growth on the real interest rate. Importantly, the theory suggests that a negative real interest rate can arise naturally when the expected economic growth is low and when the risk is high.

## 3. Reasons for the decline of the real interest rate

After reviewing the standard theory of interest rate determination, I now discuss the potential driving forces for the secular decline. I start with the factors highlighted by the canonical models and then discuss the factors beyond these models, including convenience yields of safe assets, demand for safe assets from emerging markets, monetary policy, relative price of capital, wealth and income inequality, and diverging trends of corporate saving and investment.

# 3.1 Factors highlighted by canonical models

3.1.1 Productivity. Eqn (5) suggests that a key determinant of the real interest rate is the growth rate of per capita consumption, which in turn is determined by the productivity growth Eqn (9). A natural candidate for explaining the fallen interest rates is then

productivity growth, measured by either labor productivity or total factor productivity (TFP). Consistent with the standard theory, Gordon (2012) shows that the labor productivity growth in the US has slowed down significantly since the 1970s despite a temporary rise from 1996 to 2004 driven by the Third Industrial Revolution (computers and internet). He also points out six factors that are in the process of dragging down long-term growth, including demography, education, inequality, globalization, energy/environment, and the overhang of consumer and government debt. In support of Gordon's argument, Figure 2 shows the total productivity growth rate in the US from 1948 to 2019. The 10-year moving average exhibits a noticeable drop since the mid-1970s.

However, using a long sample (1890–2016) of the US data, Lunsford and West (2019) find that in contrast to the standard theory there is a negative rather than a positive long-run correlation between the risk-free rate and productivity growth. In addition, Yi and Zhang (2017) find that the median long-run marginal product of capital in their sample of 20 countries is essentially flat since 1980s; Gomme *et al.* (2011, 2015) show that unlike the real return on treasury securities, the real return on productive capital, which is an extended version of the investment return defined in Eqn (7), has not shown a downward trend in the US in recent decades. These findings suggest that slower productivity growth is probably not the reason for the decline in the real interest rate.

*3.1.2 Demographics.* Eqn (10) from the Diamond (1965) overlapping generations model shows that a lower population growth reduces the real interest rate due to its positive effect on the capital–labor ratio. Other demographic characteristics, such as fertility and mortality rates, life expectancy, age distribution, can also affect the interest rate through their effects on saving and labor supply. Carvalho *et al.* (2016) calibrate a life-cycle model to capture the salient features of the demographic transition in developed economies. Their model reveals three effects of the increased longevity and lower population growth on real interest rates: a negative effect through higher saving for a longer retirement period; a negative effect through reduced marginal product of capital due to higher capital-to-labor ratio; and a positive effect due to lower saving of retirees relative to workers. They show that these effects



**Figure 2.** Total factor productivity growth rate

CFRI

**Note(s):** This figure plots the annual total factor productivity growth rates and its 10-year moving average in the US from 1948 to 2019.

Source(s): Downloaded from the website of the Federal Reserve Bank of San Francisco.

together reduce the equilibrium real interest rate by at least one and a half percentage points from 1990 to 2014. Lisack *et al.* (2017) reach a similar conclusion. Eggertsson *et al.* (2019) show that the reductions in fertility and mortality are the most important driving forces for the real interest rate decline between 1970 and 2015. Lunsford and West (2019) examine over 30 variables that are hypothesized to have an influence. They find that real risk-free rates are correlated as expected with demographic measures: positively correlated with labor force hours growth and negatively correlated with the proportion of 40- to 64-year-old people in the population. Most other variables considered in their study have a mixed relationship with the real rate. These results further confirm the role of demographic transition as a driver of the decline in real-free rates in recent decades.

While the effect of demographic characteristics on the interest rate has been confirmed both theoretically and empirically, there are still good reasons to be cautious about putting too much weight on demographics. Figure 3 shows the personal saving rate and the ratio of personal saving to gross private saving in the US from 1947 to 2019. There is a sharp decline in the personal saving rate from 1971 (13.5%) to 2005 (3.2%). As a result, the role of personal saving in the gross saving of the private sector is greatly diminished (its weight drops from 44% to 12%). Therefore, any potential positive effect of demographic transition on personal saving rate must have been dominated by other factors.

*3.1.3 Risk.* Eqn (5) highlights the negative effect of risk on the risk-free rate. Barro (2006) shows that the risk of rare disasters can not only explain the long-run means of real risk-free rate in the G7 countries, but also the drops of risk-free rates during the disaster episodes such as the Second World War, because those episodes are likely associated with heightened concerns for the tail risk. The sharp declines of risk-free rates witnessed during the 2008 global financial crisis and the COVID-19 pandemic provide further examples.

One puzzle about tail events is their long-lasting effects. Why does a crisis that happened more than a decade ago continue to affect asset prices today? Kozlowski *et al.* (2018, 2020) provide an explanation. They assume that agents do not know the true distribution of shocks and learn it from data nonparametrically. By definition, tail events occur rarely, and the



**Note(s):** This figure plots the personal saving rate (personal saving over disposable personal income, left axis) and the ratio of personal saving to gross private saving (right axis) in the US from 1947 to 2019.

Source(s): Retrieved from FRED, Federal Reserve Bank of St. Louis.

Causes and challenges of low interest rates

> Figure 3. Personal saving

# CFRI

commonly observed nonevent periods provide little information about the distribution at the tails. Only the rarely observed extreme events are used to adjust the belief. Therefore, the effect of those events are long-lasting. These authors show that their calibrated models can account for many macroeconomic phenomena post the 2008 financial crisis. In particular, Kozlowski *et al.* (2018) show that by incorporating time-varying convenience yields of safe assets, updated beliefs about tail risk due to the 2008 crisis can reduce the risk-free rate by as much as 145 basis points. Importantly, they show that the effect is very persistent.

A market-based tail risk index, the Chicago Board Option Exchange (CBOE) SKEW Index, which is calculated from the prices of out-of-money S&P500 options, is consistent with the tail risk–based explanation. While standard measures of market risks, such as the CBOE VIX index, revert to normal soon after the peak of the financial crisis, the SKEW index remains on an upward trajectory for another ten years, as shown in Figure 4. Its subsequent decline is again reversed by a new rare disaster: COVID-19. However, in the 18 years before the 2008 financial crisis, for which the SKEW index is available, the relation between the risk-free rate and the SKEW index appears to be rather random. One possible explanation is that the crises in the 1990s are largely confined in developing countries (for example, the Mexican, Asian, and Russian crises). These crises increase the global demand for US safe assets but are not fully reflected in the SKEW index.

# 3.2 Factors beyond canonical models

*3.2.1 Convenience yields of safe assets.* While canonical models focus on the role of safe assets in smoothing intertemporal consumption, the recent literature has highlighted another aspect of these assets: the convenience yields (Krishnamurthy and Vissing-Jorgensen (2012), Greenwood *et al.* (2015)). Due to its high liquidity and high safety, safe assets such as government bonds are used as collateral in many financial transactions and used by banks and money market funds to back checkable deposits and meet withdrawals. As a result, they perform some medium of exchange role of money. Since investors value such nonpecuniary



**Note(s):** This figure shows the real one-year Treasury rate (left axis) and the COBE SKEW index (right axis) from January 1990 (when the index was first introduced) to May 2020. Both series are 60-day moving averages of daily data.

Source(s): Downloaded from the CBOE website.

Figure 4. Tail risk and real interest rate benefits, the returns they demand from safe assets are reduced, relative to other assets offering only pecuniary payoff. Krishnamurthy and Vissing-Jorgensen (2012) estimate that convenience yields reduce the US Treasury yields by 73 basis points, on average, from 1926 to 2008.

Importantly, convenience yields of safe assets are time-varying, spiking up during financial crises, when the liquidity constraints tighten up and safe assets are in shortage (Nagel (2016) and Van Binsbergen *et al.* (2019)). This further explains why risk-free rates tend to decline during tail events. In fact, Kozlowski *et al.* (2018) show that the explanatory power of tail risk for the real interest rate fall post the 2008 financial crisis hinges on time-varying liquidity constraints.

The significant convenience yields of safe assets suggest that the observed nominal yields underestimate the true benefits of holding these assets. This can explain why nominal risk-free rates can be negative, as seen in Japan and Europe. While paper currency offers a nominal return of zero, it is not a perfect substitute for electronically stored safe assets such as government bonds and bank deposits due to its storage costs and its inconvenience as a means of payment, especially for large transactions. However, although convenience yields are useful for explaining the low level of the risk-free rate, they are less helpful for explaining the downward trend since the 1980s. Most empirical estimates of safe asset convenience yields do not show a clear upward trend that is consistent with the secular decline of the real interest rate. For example, the convenience yield on treasuries estimated by Van Binsbergen *et al.* (2019) based on option prices averages to 40 basis points over the sample period 2004–2018, and it is slightly higher in the precrisis period than in the postcrisis period, inconsistent with the lower risk-free rate post the crisis [6].

3.2.2 Demand from emerging markets. Bernanke (2005) argues that the large current account deficits and low interest rates in the US are caused by a "global saving glut," pointing to the increasingly strong demand for safe assets of developed countries from emerging markets since the 1990s. This somewhat surprising pattern of international capital flows is partly due to a series of financial crises in emerging markets in the 1990s, which not only raise the tail risk concerns of international investors but also push central banks in emerging markets toward accumulating more foreign reserves to better prepare for future crises.

Caballero *et al.* (2008) present a model of global imbalance in trades and capital flows. They consider the cross-country heterogeneities in the capacity to generate financial assets from real investments and in saving rate. Due to the underdevelopment of financial markets, countries like China have lower capacity to capitalize future incomes and transform them into tradable assets. The limited supply of saving assets lowers the interest rate. When such countries integrate to the world economy, global demand for financial assets of advanced economies increases, leading to a large current account deficit in these economies and a decline of the world interest rate. The integration of countries with a high saving rate into the world economy has a similar effect. Since countries with underdeveloped financial markets and high saving rates also tend to grow faster, the world interest rate declines over time as these countries gain more weight in the world economy.

Hall (2017b) proposes a similar explanation for the secular decline of the risk-free rate but with a focus on heterogeneities in risk aversion and beliefs. He assumes that some countries, such as China and Japan, have higher risk aversion and more cautious beliefs about the probability of rare disasters. As the wealth of these countries grows relative to the more risk-tolerant countries, such as the US, the demand for safe assets grows, and the world-wide risk-free rate declines.

It is certainly true that interest rates are increasingly driven by global factors due to the integration of the world economy. However, the impact of demand from emerging markets is hard to quantify. While industrial countries as a whole have net financial capital inflows in recent decades, they also have large net FDI (foreign direct investment) outflows. Wang *et al.* (2017) shows that when the two-way capital flows are considered simultaneously, the impact

of capital flows on the world interest rate can be quantitatively negligible. Furthermore, among the countries with the world's lowest interest rates, some enjoy persistent current account surpluses (e.g. Germany and Japan). Clearly, low interest rates in those countries cannot be attributed to the global imbalance.

3.2.3 Monetary policy. One common feature of the canonical models reviewed in Section 2 is that they all focus on the real determinants of real interest rates and ignore the role of monetary policy. This is consistent with the notion of classical dichotomy - real variables are determined by real factors and nominal variables are determined by monetary policy. According to this view, the effect of monetary policy on real interest rates is short-lived at most. Such a dichotomy may be too simplistic. Empirical evidence surveyed by Williams (2014) and Kuttner (2018) suggests that large-scale purchases of long-term securities by central banks have sizable effects on long-term interest rates. In particular, Krishnamurthy and Vissing-Jorgensen (2011) find that the quantitative easing operations conducted by the Federal Reserve not only lowered the nominal long-term risk-free rates but also increased expected inflation, implying a larger reduction in real rates than in nominal rates. In addition to quantitative easing, the tightened postcrisis financial regulations have also increased demand for safe and liquid assets, adding further pressure on the risk-free rates. However, since the decline in real interest rate occurred long before the Great Recession, any explanation based on the postcrisis monetary policy is unlikely to be a full story. Furthermore, while monetary authorities can determine the nominal rate, they have no ability to keep the real rate substantially above or below the natural rate persistently. Therefore, the role of monetary policy in the secular real interest rate decline should not be exaggerated.

3.2.4 Relative price of capital. Together with the decline of the real interest rate is a steady decline in the price of investment goods relative to consumption goods as well as the decline of the investment-output ratio. Using a large global sample, Karabarbounis and Neiman (2014) show that the relative price of investment goods fell by almost 30% between the late 1970 and 2000s. Sajedi and Thwaites (2016) show that the nominal investment-to-output ratio declined by about one-quarter between 1980 and 2012. Summers (2014) argues that a lower relative price of capital should have a negative effect on the interest rate because it means that the same investment projects can be funded by less savings. Even though the lower price would encourage firms to expand investment to less profitable projects, when the elasticity of substitution between capital and labor is less than one, the total spending on capital still falls, resulting in a lower investment-out ratio and a lower interest rate. Sajedi and Thwaites (2016) explore this possibility in an overlapping generations model. In response to a 25% drop in the relative price of capital, their calibrated model can reproduce about one-third of the empirically observed fall in the investment-output ratio. However, it generates a fall in the real interest rate only by 12 basis points, suggesting that the relative price of capital is unlikely to be an important driver of the real interest rate decrease. Furthermore, Eichengreen (2015) shows that the quality-adjusted relative price of investment has been falling steadily since the early 1950s, long before the start of the real interest rate decline. This casts further doubt about the explaining power of the falling relative price of capital.

3.2.5 Wealth and income distribution. Another potential contributing factor that has drawn a lot of attention is the rise of inequality in income and wealth distributions. There is a substantial body of evidence showing the decline of the labor share in corporate gross value added since the 1970s. For example, Karabarbounis and Neiman (2014) find that the global labor share drops from 61% in 1975 to 59% in 2012 [7]. In addition, income has become increasing skewed toward the top 1% or 0.1% (see, for example, Atkinson *et al.* (2011)). While the diminished labor share and increased income/wealth inequality can in principle have both positive and negative effects on real risk-free rate, several recent studies have shown that the negative effect turns to dominate under reasonable model calibrations (Favilukis (2013), Kaymak, and Poschke (2016); Eggertsson *et al.* (2019)), due to the higher saving rate of the

richer and the higher precautionary saving associated with more dispersed income shocks [8]. However, the estimated magnitudes of these effects are typically relatively small (for example, Eggertsson *et al.* (2019) estimate the effect of the labor share decline from 1970 to 2015 on the real interest rate to be -52 basis points).

3.2.6 Corporate investment and saving. One potentially important contributing factor to the real interest rate decline, which has not been fully explored in the literature, is the diverging trends of corporate saving and investment. Using a large international sample, Chen *et al.* (2017) document a global rise in corporate saving, measured as undistributed corporate profit. Consistent with what Figure 3 shows for the US, they also document a pervasive shift in the composition of saving away from the household sector and toward the corporate sector. While most of global investment was funded by household saving in the early 1980s, nearly two-thirds of it is funded by corporate saving in 2013.

Following the definition of corporate saving and investment in Chen *et al.* (2017), I plot in panel (a) of Figure 5 the ratio of corporate sector aggregate investment to aggregate saving in the US from 1980 to 2019, constructed using the Compustat North America database [9]. Although the ratio is quite volatile, the overall downward trend is obvious, suggesting that firms rely less on outside finance for capital formation in more recent years. Chen *et al.* (2017) identify the low interest rate as one of the reasons for the dramatic increase in corporate saving. However, the causality can also go in the opposite direction: given the size of the corporate sector, rising corporate saving relative to investment can be an important reason for the decline of the interest rate.

Consistent with the declining investment-saving ratio of the corporate sector, there is a dramatic increase in corporate cash holdings since the 1980s. Bates *et al.* (2009) document that the average cash-to-asset ratio increased from 10.5% in 1980 to 23.2% in 2006. The cash hoarding is especially prevalent among the superstar firms such as Google, Microsoft and Apple, leading to a large increase in aggregate. Panel (b) of Figure 5 shows the times series of the ratios of three different components of aggregate corporate investment, i.e. capital expenditures, R&D, and acquisition expenditures to cash are relatively stable, but the ratio of capital expenditures to cash drops by more than three-quarters, from 1.85 in 1980 to 0.43 in 2019. This dramatic drop is due to both the decline of capital expenditures and the increase of cash holdings: from 1980 to 2019, aggregate capital expenditures scaled by lagged capital stock (measured by net value of property, plant and equipment) drop from 24% to 15%, while aggregate cash holdings scaled by lagged capital stock increase from 14% to 37%. At the end of the fiscal year 2019, the total amount of cash held by the firms in the sample amounts to \$3.4 trillion [10].

There are several reasons for the fallen investment-saving ratio. First, the increase of firms' market power, which is most evidently shown by the rise of superstar firms in many industries (Autor *et al.* (2020)). Market power not only increases firms' profitability and valuation but also creates a gap between the average Tobin's *Q* and the marginal *Q*, weakening firms' incentive to invest, as shown by Gutierrez and Philippon (2017); Farhi and Gourio (2019) and Corhay *et al.* (2020). Second, changes in economic environments, including more global tax arbitrage opportunities, more disruptive innovations, larger first mover advantages and higher asset intangibility, induce firms to hold more cash instead of paying it out to shareholders; at the same time, a lower interest rate also lowers the cost of carrying cash [11]. Third, a misallocation of financial resources and investment opportunities. The established firms flooded with cash may not have the best investment opportunities or the incentives to engage in radical innovations, while those with opportunities and right incentives have difficulty in raising capital [12].







**Note(s):** Panel (a) shows ratio of aggregate corporate investment to saving. Panel (b) shows the ratios of aggregate capital expenditures (CAPX), R&D, and acquisition expenditures to lagged aggregate cash holdings. Investment is defined as sum of capital expenditures (net of sales of property, plant & equipment), R&D expenses, and acquisition expenditures; saving is defined as operating income before depreciation (OIBDP in Compustat) plus R&D, minus taxes, interests, dividends, and share repurchases. The sample includes all nonfinancial, non utility firms covered in the Compustat North America database from 1980 to 2019. Firms are required to file financial reports in the US dollar. In panel (b), they are also required to have positive cash holdings at the prior fiscal year end.

Source(s): Compusta North America.

Figure 5. Corporate investment, saving and cash holdings Given the central role of the corporate sector in capital formation and technological developments and its increasingly important role in funding these activities, a better microlevel understanding of what drives the diverging trends of corporate saving and investment and their macro implications is crucial for understanding the current low interest rates and potential inefficiencies in the economy.

To summarize, there has been very active research on the causes of the secular decline of real interest rates, and significant progresses have been made. Each explanation offered in the literature is helpful for understanding this trend. However, none of them fully accounts for the observed interest rate dynamics. It is plausible that the decline is a joint outcome of most if not all the factors discussed above. A fruitful area for future research is the changes in industry structure that alter the investment and saving behaviors of the corporate sector. To the extent that most of economic forces driving the downward trend represent relatively slow evolution of the real economy without a clear sign of reversal in the near future, low real risk-free interest rates are likely to persist for some considerable time.

## 4. Challenges for monetary policy

The extended period of low real interest rates and low inflation have posed serious challenges for conventional monetary policy. It also challenges some conventional wisdoms of monetary theory and sparks radically new ideas. As Rogoff (2017) writes, "despite an outward appearance of stability, the core of the global monetary system today is immersed in a level of intellectual turmoil not seen since the breakup of the Bretton Woods system in the early 1970s". In this section, I briefly discuss the monetary policy conundrum in the low interest rate environment and the policy implications of two recent developments in monetary theory, the neo-Fisherism and the fiscal theory of price level.

#### 4.1 The monetary policy conundrum at the zero bound

Because of the unstable relation between money supply and price level, starting from the 1980s, most central banks abandoned money growth as the main instrument for monetary control. Instead, they have relied heavily on a "leaning-against-the-wind" interest rate policy. The famous Taylor rule specifies how nominal interest rate  $i_t$  should respond to divergences of actual inflations  $\pi_t$  from target inflation rate  $\pi_t^*$  and of actual GDP  $y_t$  from potential GDP  $\overline{y}_t$ :

$$\dot{i}_t = \pi_t + r_t^* + a_\pi (\pi_t - \pi_t^*) + a_y (y_t - y_t),$$

where  $r^*$  is the equilibrium real interest rate (the natural rate), and  $a_\pi > 0$  and  $a_y > 0$  describe the responses to inflation and output deviations. The rule thus says that central banks should reduce the nominal interest rate, usually through open market purchases of short-term government bonds, when inflation and GDP are above their target levels, and vice versa. In his original paper, Taylor (1993) set the target inflation  $\pi_i^*$  at 2%, although he provided little discussion for this choice. This number has been embraced by many central banks in advanced economies (including the USA and Japan), potentially because it strikes a balance between the cost of inflation and the risk of deflation. Central banks in developing countries tend to have a slightly higher target inflation, but mostly in single digits.

When the natural rate is 3%, a target inflation rate of 2% means a nominal interest rate of 5% in normal times. Assuming that the nominal interest rate has a lower bound at zero (which is referred to as the zero bound), a nominal rate of 5% allows the central bank to cut the interest rate by a maximum 5 percentage points. However, when the natural rate is 1%, the nominal interest is only 3% in normal times. According to Summers (2018), the average rate cut implemented by the Federal Reserve since 1960s to combat recessions is 5%. Therefore, there is a concern about not having enough room for rate cuts when the next recession arrives. Even

worse, since the 2008 financial crisis, the Fed has constantly fallen short of the 2% target inflation rate, and the short-term nominal interest rate stays close to zero for most of the time.

Of course, the Taylor rule is not the only tool at central banks' disposal. During the Great Recession and the subsequent low interest rate years, the Federal Reserve System intervenes through so-called quantitative easing, which involves large-scale purchases of public or private debt of longer maturities. Such operations allow the Fed to lower interest rates at longer maturities and target at specific market sectors (such as the mortgage markets). It also actively uses "forward guidance" to influence expectation about the future path of interest rates. Central banks in Europe and Japan go beyond the zero bound to experiment with negative nominal rates. Upon the outbreak of the COVID-19 crisis, the central banks react swiftly, displaying a whole spectrum of weapons they command. For example, in addition to decisive rate cuts and massive purchases of long-term securities, the Fed lends directly to banks, primary dealers, corporate employers and state and local governments. It also provides liquidity support to money market funds, repo markets, commercial paper markets and temporarily relaxes bank regulatory requirements.

Nevertheless, these unconventional operations have their own limitations, and people are rightly concerned about their potential side effects. Therefore, many economists have argued for raising the target inflation rate to get more headroom for the interest rate policy. For example, Summers (2018) argues for a monetary framework that would "foresee nominal interest rates in the range of 5 percent in normal times." Williams (2016) expresses similar views [13]. However, even if a switch to a higher inflation target is desirable, central banks still face the challenge of how to achieve the target, as many of them, especially Bank of Japan, have realized, generating inflation is not always an easy task.

Rogoff (2017), on the other hand, is less concerned about the zero bound and advocates for clearing the way to allow normal interest rates to go deeper in the negative territory. Pointing to the current practice in Japan and Europe, he argues that, fundamentally, there is no practical obstacle to paying negative interest rates on electronic currency. The main concern is that if a central bank pushes the interest rate on electronic currency too deeply negative, there will a massive flight into paper currency. However, in today's world, the use of paper currency as a means of payment has been largely limited to small transactions. Even for those transactions, electronic payments by credit cards or smartphones have become increasingly ubiquitous. The recent launch of a digital currency by the central bank of China is the latest example of this trend. As a result, the issue of a potential flight to paper currency may not be as hard to address as it used to be. Rogoff (2017) proposes two approaches. One involves getting rid of large-denomination notes, and the other involves creating a crawling pegged exchange rate between paper currency and bank reserves.

However, even if a significantly negative nominal interest rate is operationally feasible, its stimulating effect is not guaranteed. Brunnermeier and Koby (2019) show theoretically that there is a threshold beyond which further interest rate cuts will become contractionary. They call this threshold rate the reversal rate. The reversal occurs because of two opposing effects of a rate cut on banks' net worth. On the one hand, a rate cut generates capital gains on banks' long-term fixed-rate assets. On the other hand, it shrinks banks' net interest income going forward. When the second effect dominates, a rate cut reduces banks' lending capacity and becomes contractionary. The exact level of this reversal rate depends on banks' initial asset holdings and capitalization, the strictness of capital requirement, and the degree of the policy rate pass-through to deposit rates. Importantly, the reversal rate is not necessarily zero. In fact, the structural estimation conducted by Wang *et al.* (2019) shows that when the federal fund rate falls below 0.9%, the money policy effect will be reversed due to the interaction between banks' market power and capital requirement. According to this estimate, the current interest rate at the zero bound is too low to be stimulating. Furthermore, Acharya *et al.* (2019) find that the glut of cheap credit has a disinflationary effect by allowing many

struggling "zombie firms" to stay afloat, which creates excess production capacity and downward pressure on product prices.

## 4.2 Neo-Fisherism

The low interest rate policy as well as the whole philosophy of the Taylor rule are also criticized by a new school of the monetary theory: the neo-Fisherism. The famous Fisher Effect, hypothesized by Irvine Fisher in his classic book *Theory of Interest* published in 1930, describes the following relationship between nominal interest rate  $R_t$  and real interest rate  $r_i$ :

$$R_t \approx r_t + E_t(\pi_{t+1}),$$

where  $E_t(\pi_{t+1})$  is the expected inflation rate. This hypothesis states that nominal interest rates tend to move in parallel with inflation rates so that monetary policy has little effect on the real interest rate, a version of the classical dichotomy (or money neutrality).

Empirically, there is a robust relation between nominal interest rates and inflation: high inflation is accompanied by high nominal interest rates, both over time and across countries (see, for example, Williamson (2019)). The conventional interpretation of this relation is that high inflation causes high nominal interest rates: because both borrowers and lenders care about the real rate, when they expect high inflation, they would mutually agree to set a high nominal interest rate. The neo-Fisherism departs from this conventional wisdom with two radical claims: (1) it turns the causality around and argues that it runs from the nominal interest rate targeted by the central bank to inflation, (2) it argues that a permanent increase in the nominal interest rate leads to an increase in inflation not only in the long run but also in the short run. These claims are supported by a number of theoretical analysis including Schmitt-Grohe and Uribe (2014); Cochrane (2016), and Williamson (2018, 2019), as well as some empirical evidence Williamson (2019), and Uribe (2019).

The Neo-Fisherian view of monetary policy questions the validity of the Taylor rule as a principle for inflation control. According to the Fisher equation, for an increase in the nominal interest rate to induce a decrease in anticipated inflation the real interest rate must increase more than one-for-one with the nominal interest rate. This requires a large money nonneutrality and thus may be unrealistic. Neo-Fisherians further argue that the conventional practice may be the reason why many central banks find themselves trapped at the zero bound: When inflation is falling, a central bank following the Taylor rule reacts by cutting the nominal interest rate. However, because of the Fisher effect, this actually leads to lower inflation, which causes further nominal interest rate cuts. Ultimately, the nominal interest rate hits the zero lower bound. This tendency of converging to the liquidity trap is referred to as "the Perils of Taylor Rules" (Benhabib *et al.* (2001)).

The Neo-Fisherian policy recommendation for the exit from the liquidity trap at the zero bound is both simple and radical: instead of a negative interest rate or quantitative easing, central banks should increase the nominal interest rate, and a higher inflation will follow (Grohe and Uribe (2014), Williamson (2016, 2019)). Whether this radical view can gain traction among central banks remains an open question. The seemingly simple solution is unlikely to be a free lunch, unless the interest rate is already below the reversal rate. Theoretical analysis by Cochrane (2020) shows that there is a temporary output fall in response to a permanent nominal interest rate increase, even if inflation starts to rise immediately. If inflation drops initially before it rises, the negative output response is even more severe [14].

### 4.3 The fiscal theory of price level

To Cochrane (2018), the post-crisis low interest/low inflation period is "a decisive experiment" that overturns many conventional wisdoms. The prolonged situation of massive increases in

bank reserves on one hand and stable and below-target inflation rates on the other is inconsistent with the traditional monetarist quantity theory of price level, which postulates that price level P is determined by money supply M: MV = PY, where V is more or less constant and Y is the output. It also challenges the Keynesian view of a need for a responsive interest rate policy such as the Taylor rule to keep the inflation stable. Cochrane (2018) argues that the recent macroeconomic experience can be best explained by the fiscal theory of price level. First developed in 1990s, this alternative monetary theory has gained more momentum recently. It incorporates some elements of the neo-Fisherism but aims to provide a more general framework for macroeconomic analysis. The main contributors include John Cochrane, Eric Leeper, Christopher Sims and Michael Woodford. Cochrane (2020) provides a systematic exposition of the theory.

According to the fiscal theory of price level, inflation is fundamentally anchored by fiscal policy instead of money supply. The value of money comes from the government's commitment to accept it for tax payments. Nominal government debt represents the claim to future primary fiscal surpluses (tax incomes minus government spending excluding interest payments), just as a stock represents a claim to a company's stream of dividends. The real value of government debt, which is the nominal value divided by the price level, must equal the present value of primary surpluses. When the amount of nominal debt is fixed, the price level moves inversely with the real debt value, which in turn is determined by the expected stream of primary surpluses and the real discount rate. If the expected stream of surpluses falls or the real discount rate increases, the discounted value of surpluses falls, and the price level must increase until the real value of the debt is equal to the lower discounted value. Economically, this mechanism materializes as follows: if people feel that fiscal surpluses are insufficient to pay off the government debt or that the return from the government debt is too low relative to the required return (i.e. the discount rate), they will sell government debt in exchange for goods and services, which pushes up the price level. Thus, by making the price level adjust endogenously to satisfy the valuation equation for government debt, the fiscal theory offers an alternative explanation for the determination of price level, rooted in the supply and demand of government debt.

This theory offers a radically new explanation for the low inflation since the 2008 financial crisis. Both nominal and real interest rate dropped sharply in 2008, driven by poor growth perspectives and a flight to quality. For reasons reviewed in Section 3, the downward trend continues even after the crisis. Therefore, despite the large increase in government deficit, the discounted real value of government debt still increases. To match this increase in real value of government debt, the price level fell during the crisis and increased little since then.

The theory also offers a new perspective on the current macroeconomic situation and monetary policy choice. First, there is no need to worry unduly about the zero bound. It is possible to have a stable economy with a low nominal interest rate insensitive to economic conditions. An active interest rate policy is not necessary for inflation stability or determinacy. Second, there is no need to worry unduly about the inflationary effect of large bank reserves. From the banks' perspective, bank reserves issued by central banks in exchange for government debt are almost perfect substitutes for the government debt itself. Therefore, banks holding reserves are not more inflationary than banks holding government debt, which explains why quantitative easing has no effect on inflation. Third, while central banks *can* maintain a fixed nominal interest rate and let inflation adjust to variation in real interest rate, they can also respond more actively to variation in real interest rates to reduce the necessary adjustment in inflation, which may otherwise take a long time. The theory also suggests a good target for such an active policy: the interest spread between nonindexed debt and indexed debt (such as the TIPS), which reflects expected inflation. By focusing on the spread instead of the interest rate level, this strategy nails down expected inflation but allows the interest rate level to adjust automatically to market forces. Unlike the Taylor rule, aiming

at this target does not require an estimation of the natural interest rate. Finally, price stability can only be achieved on the foundation of fiscal stability. Once the market loses confidence in the government's ability to repay its debt, the real value of government debt drops and inflation may be out of control.

Given that the majority of money in today's economy is not "chasing goods," but chasing stocks, bonds, real estate and other assets, it is not surprising that the quantity of money is no longer linked so directly to the CPI or other price indexes of goods and services. An alternative theory of price level based on the supply and demand of government debt offers a valuable new perspective. However, the valuation of government debt proves to be challenging. Jiang et al. (2020) estimate a quantitative model of the US government debt and find a large gap of the size of 287% GDP between the value of the aggregate surplus claim and the market value of the outstanding debt. Because primary fiscal surpluses are strongly procyclical, the standard asset pricing theory implies that the aggregate claim to fiscal surpluses should earn a risk premium, but investors seem to be willing to purchase government debt at low yields. Since the valuation of government debt plays a central role in the fiscal theory of price level, the lack of a clear understanding of how it works is unsettling. Furthermore, while the fiscal theory of price level seems to provide a more optimistic view of the economic situation near the zero bound, it is important to recognize that the stability we observe is still just an experience of limited time, and it is conditioned on massive holdings of government debt by central banks. The subjectivity of government debt discount rate implies that it could increase abruptly, in which case price stability will be jeopardized.

Even if the macroeconomic stability is not at immediate risk, low interest rates still bring many pains, especially for financial institutions and populations that rely heavily on stable cash flows from fixed-income assets, such as banks, insurance companies, endowments and retirees. Take the two largest German banks, Deutsch Bank and Commerzbank, as examples. By the end of May 2020, Deutsch Bank stock price has dropped by 88% since 2010, following a more than 50% drop during the 2008 financial crisis; Commerzbank stock price has dropped by 94%, following a more than 80% drop during the financial crisis. A fragile banking sector is hardly conducive to a dynamic economy. It can itself be a source of instability. In addition, there are also concerns about the implications of low interest rates on long-term growth. For example, Liu *et al.* (2019) show that a low interest rate gives industry leaders a strategic advantage over followers, which in turn reduces competition, investment and overall productivity growth.

# 5. Conclusion

It is now time to conclude this somewhat long intellectual journey in the theory and reality of the real interest rate. What are the main takeaways from this journey?

The neoclassical asset pricing theory and growth theory provide a useful framework for understanding the downward trend of real interest rates. The canonical models suggest that the risk-free interest rate is determined by the subjective discount factor, the expected consumption/productivity growth, the riskiness of the consumption/productivity growth, and the population growth. It also suggests that a negative real interest rate can arise naturally when the expected growth rate is low and when the risk is large. These predictions are helpful for understanding the downward trend.

The decline coincides with a slowdown of productivity growth since the 1970s. However, recent studies only find a tenuous long-run correlation between the real risk-free rate and productivity growth. In addition, the real return on productive assets has been stable in recent decades. These findings suggest that slow productivity growth may not be the reason for the secular decline of the real interest rate. The negative effects of some demographic trends, including lower population growth, lower fertility rate, increased longevity, have received

stronger empirical support. Nevertheless, the quantitative importance of these demographic trends needs to be assessed with caution, given that these trends coincide with a sharp decline in the personal saving rate and a diminished share of households' contribution to the aggregate saving of the economy. Elevated tail-risk concerns, together with large convenience yields of government bonds during extreme events, provide a credible explanation for the decline of real interest rate after the 2008 financial crisis. Since agents do not know the true shock distribution, extreme events such as the 2008 crisis and the 2020 pandemic can have a long-lasting effect on the perceived tail-event probability. Tail-risk concerns may also have contributed to the increased demand for safe assets of developed countries after the financial crises in emerging markets in the 1990s.

Factors beyond the standard theory also help to explain the interest rate decline. These include the increased global demand for safe assets of developed countries, the accommodative monetary policy, the falling relative price of capital, and the rising wealth/ income inequality. The convenience yields of safe assets provide an explanation for negative nominal risk-free rates in Japan and Europe. Given the large increase in corporate saving relative to corporate investment in recent decades, a better understanding of economic forces driving the diverging trends of corporate investment and saving behaviors should be very helpful for understanding the trend of the real interest rate.

A critical review of each explanation put forward in the literature suggests that each explanation has its merits, but none of them provides a full qualitative and quantitative account of the real interest rate dynamics in recent decades. It is plausible that the secular decline of the interest rate is a joint outcome of most if not all the factors mentioned above. Furthermore, since most of these factors represent relatively slow evolution of the real economy without a clear sign of reversal in the near future, the low real risk-free rate is likely to persist for some considerable time.

The low interest rate poses a conundrum for monetary policy as it limits the space for rate cuts in response to potential crises. Even if a negative nominal rate is theoretically justifiable and technically feasible, it is not necessarily stimulating. Due to its negative impact on banks' net worth, a low interest rate can reverse to become contractionary. Importantly, this reversal rate is not necessarily negative.

The low interest rate and low inflation economy also push economists to rethink the fundamental principles of monetary theory. The neo-Fisherist view of the monetary policy argues that the conventional Taylor rule–based policy has a tendency of falling into the liquidity trap. A radical policy recommendation coming out of this theory is that in order to break the liquidity trap central banks should increase instead of decreasing the nominal interest rate. If the prevailing interest rate is below the reversal rate, then a rate increase can indeed be stimulating. Unconventional monetary policies like this should be evaluated with caution as there may be unintended consequences. Quantitative estimation of the reversal rate such as done by Wang *et al.* (2019) is very useful for this purpose.

According to the fiscal theory of price level, the general price level is determined by the demand and supply of government debt instead of money supply. Quantitative easing itself has little inflationary effect because government debt and bank reserves are almost perfect substitutes for banks. Furthermore, it is possible to have a stable economy with a low interest rate without a Taylor rule-type intervention policy. Given that the majority of money in today's economy is used for financial transactions instead of buying goods and services, an alternative theory of price level based on the supply and demand of government debt offers a valuable new perspective. However, it is important to recognize the negative effects of low interest rates on institutions and populations relying on stable cash flows of fixed-income assets. Also, the valuation of the aggregate government debt portfolio is still a challenge for the asset pricing theory. The subjectivity of government debt discount rate implies that it could increase abruptly when investors become concerned about the government's ability to pay off its debt, in which case the resulting runaway from government debt in exchange for

goods and services will jeopardize price stability. The fiscal theory of price level builds on the idea that price level is anchored by fiscal policy; a natural conclusion that should follow is that price stability must be built on fiscal discipline.

# Notes

- 1. Instead of measuring the real interest rate as the nominal rate minus the *expected* inflation rate, one can also measure it as the difference between the nominal rate and the *realized* inflation rate. Using this ex post measure, the peak real interest rate in the early 1980s is over 12%. Therefore the subsequent decline is even more dramatic. Since a forward-looking measure is more meaningful for economic decisions, the real interest rate in this survey always refers to the ex ante real rate.
- 2. A lognormally distributed random variable X has the following convenient property:

$$\ln E(X) = E[\ln(X)] + \frac{1}{2}\sigma^{2}[\ln(X)],$$

where  $\sigma^2[\ln(X)]$  denotes the variance of  $\ln(X)$ .

- 3. The intuition for a higher sensitivity of the risk-free rate to the expected consumption growth rate is as follows. Under the power utility,  $\gamma$  not only measures the aversion against variation of consumption across states (i.e. risk) but also measures the aversion against variation of consumption over time ( $1/\gamma$  is equal to the elasticity of intertemporal substitution). A higher  $\gamma$  implies a stronger desire to smooth consumption over time, which means that a larger move in the interest rate is needed for the agent to accept a certain expected change in consumption.
- 4. Under the power utility, the equity premium puzzle and the risk-free rate puzzle are the two sides of the same coin because of the mechanical inverse relation between risk aversion and elasticity of intertemporal substitution. However, Weil (1989) shows, using the now famous Epstein–Zin–Weil preferences, that both puzzles remain intact when these two preference parameters can be calibrated independently.
- 5. This result is easily obtained by extending the canonical overlapping generation model in Chapter 9 of Acemoglu (2009) to allow for depreciation and labor-augmenting productivity growth.
- Del Negro *et al.* (2017) estimate an increase of convenience yield on treasuries by 93 basis points from 1998 to 2016.
- 7. See also Caballero et al. (2017) and Greenwald et al. (2020).
- 8. Eqn (10) shows that the interest rate is positively related to the capital income share  $\alpha$  in the canonical growth model with overlapping generations. This is because saving in the model comes exclusively from the labor income of the young generation. The old generation owns all capital and consumes all capital income. There is no heterogeneity within a generation.
- 9. My definition of corporate saving differs from theirs in one aspect: I view share repurchase as an alternative way of paying dividends and exclude it from what they define as corporate saving. Without this modification the investment-saving ratio declines even more.
- 10. Following the standard practice in corporate finance research, cash is measured as the sum of the balance sheet items "cash and cash equivalents" and "short-term investments." Duchin *et al.* (2017) find that on average this measure underestimates firms' financial asset holdings by 25% because some financial assets are reported as "long-term investments" and "other assets." They also show that 61.7% of the aggregate financial asset portfolio is invested in money-like safe assets.
- 11. Fritz *et al.* (2007); Azar *et al.* (2016); Mello *et al.* (2020) and Falato *et al.* (2020) analyze how these factors increase corporate cash holdings.
- Dangl and Wu (2016) provide an explanation for slow recoveries of corporate investment after recessions.
- 13. On August 27, 2020, Fed Chairman Powell (2020) announced a major shift in the Fed's approach in managing inflation. Rather than making 2% a fixed goal, the Fed now seeks to achieve inflation that

averages 2% over time. This more flexible inflation target means that the Fed may aim to achieve inflation moderately above 2% for some time following periods of inflation below 2%.

14. However, empirical estimation by Uribe (2019) shows that while a temporary increase in nominal interest rate decreases inflation and output and increases real rates, a permanent nominal interest rate increase leads to an immediate increase in inflation and output and a decline in real rates.

# References

- Acemoglu, D. (2009), Introduction to Modern Economic Growth, Princeton University Press, Princeton, NJ.
- Acharya, V.V., Matteo Crosignani, T.E. and Christian, E. (2019), "Zombie credit and (dis-)inflation: evidence from europe", New York University Working Paper.
- Atkinson, A.B., Piketty, T. and Saez, E. (2011), "Top incomes in the long run of history", *Journal of Economic Literature*, Vol. 49 No. 1, pp. 3-71.
- Autor, D., Dorn, D., Katz, L.F., Patterson, C. and Van Reenen, J. (2020), "The fall of the labor share and the rise of superstar firms", *Quarterly Journal of Economics*, Vol. 135 No. 2, pp. 645-709.
- Azar, J.A., Kagy, J.F. and Schmalz, M.C. (2016), "Can changes in the cost of carry explain the dynamics of corporate 'cash' holdings?", *Review of Financial Studies*, Vol. 29 No. 8, pp. 2194-2240.
- Barro, R.J. (2006), "Rare disasters and asset markets in the twentieth century", Quarterly Journal of Economics, Vol. 121 No. 3, pp. 823-866.
- Bates, T.W., Kahle, K.M. and Stulz, R.M. (2009), "Why do US firms hold so much more cash than they used to?", *Journal of Finance*, Vol. 64, pp. 1985-2021.
- Benhabib, J., Schmitt-Grohé, S. and Uribe, M. (2001), "The perils of taylor rules", *Journal of Economic Theory*, Vol. 96 Nos 1-2, pp. 40-69.
- Bernanke, B.S. (2005), *The Global Savings Glut and the US Current Account Deficit. Remarks at the Sandridge Lecture*, Virginia Association of Economists, Richmond, Virginia, March 10.
- Breeden, D.T. (1979), "An intertemporal asset pricing model with stochastic consumption and investment opportunities", *Journal of Financial Economics*, Vol. 7, pp. 265-296.
- Brunnermeier, M.K. and Koby, Y. (2019), "The reversal interest rate", Princeton University Working Paper.
- Caballero, R.J., Emmanuel, F. and Pierre-Olivier, G. (2008), "An equilibrium model of global imbalances and low interest rates", *American Economic Review*, Vol. 98 No. 1, pp. 358-93.
- Caballero, R.J., Emmanuel, F. and Pierre-Olivier, G. (2017), "Rents, technical change, and risk premia: accounting for secular trends in interest rates, returns on capital, earning yields, and factor shares", *American Economic Review*, Vol. 107 No. 5, pp. 614-620.
- Carvalho, C., Ferrero, A. and Nechio, F. (2016), "Demographics and real interest rates: inspecting the mechanism", *European Economic Review*, Vol. 88, pp. 208-226.
- Cass, D. (1965), "Optimum growth in an aggregative model of capital accumulation", *Review of Economic Studies*, Vol. 32, pp. 233-240.
- Chen, P., Loukas, K. and Brent, N. (2017), "The global rise of corporate saving", *Journal of Monetary Economics*, Vol. 89, pp. 1-19.
- Cochrane, J. (1991), "Production-based asset pricing and the link between stock returns and economic fluctuations", *Journal of Finance*, Vol. 46, pp. 209-237.
- Cochrane, J. (2005), Asset Pricing, Revised Version, Princeton University Press, Princeton, NJ.
- Cochrane, J. (2016), "Do higher interest rates raise or lower inflation?", Hoover Institution Working Paper.
- Cochrane, J. (2018), "Michelson-morley, Fisher, and occam: the radical implications of stable quiet inflation at the zero bound", NBER Macroeconomics Annual, Vol. 32 No. 1, pp. 113-226.

Cochrane, J. (2020), *The Fiscal Theory of the Price Level*, Book Manuscript, available at: https://www.johnhcochrane.com/research-all/the-fiscal-theory-of-the-price-level-1.

Corhay, A., Kung, H. and Schmid, L. (2020), "Q: risk, rents, or growth?", University of Toronto Working Paper.

- Council of Economic Advisors (2015), *Long-Term Interest Rates: A Survey*, White Paper, available at: https://obamawhitehouse.archives.gov/sites/default/files/docs/interest\_rate\_report\_final.pdf.
- Dangl, T. and Wu, Y. (2016), "Corporate investment over the business cycle", *Review of Finance*, Vol. 20 No. 1, pp. 337-371.
- Del Negro, M., Giannone, D., Giannoni, M.P. and Tambalotti, A. (2017), "Safety, liquidity, and the natural rate of interest", *Brookings Papers on Economic Activity*, Spring 2017, pp. 235-316.
- Diamond, P.A. (1965), "National debt in a neoclassical growth model", *American Economic Review*, Vol. 55, pp. 1126-1150.
- Duchin, R., Gilbert, T., Harford, J. and Hrdlicka, C. (2017), "Precautionary savings with risky assets: when cash is not cash", *Journal of Finance*, Vol. 72, pp. 793-852.
- Eggertsson, G.B., Mehrotra, N.R. and Robbins, J.A. (2019), "A model of secular stagnation: theory and quantitative evaluation", *American Economic Journal: Macroeconomics*, Vol. 11 No. 1, pp. 1-48.
- Eichengreen, B. (2015), "Secular stagnation: the long view", American Economic Review: Papers and Proceedings, Vol. 105 No. 5, pp. 66-70.
- Falato, A., Kadyrzhanova, D., Sim, J. and Steri, R. (2020), "Rising intangible capital, shrinking debt capacity, and the US corporate savings glut", Board of Governors of the Federal Reserve System Working Paper.
- Farhi, E. and Gourio, F. (2019), "Accounting for maro-finance trends: market power, intangibles, and risk premia", NBER Working Paper 25282.
- Favilukis, J. (2013), "Inequality, stock market participation, and the equity premium", Journal of Financial Economics, Vol. 107 No. 3, pp. 740-759.
- Foley, C., Fritz, J.C., Hartzell, S.T. and Twite, G. (2007), "Why do firms hold so much cash? A tax-based explanation", *Journal of Financial Economics*, Vol. 86 No. 3, pp. 579-607.
- Gomme, P., Ravikumar, B. and Rupert, P. (2011), "The return to capital and the business cycle", *Review of Economic Dynamics*, Vol. 14 No. 2, pp. 262-278.
- Gomme, P., Ravikumar, B. and Rupert, P. (2015), Secular Stagnation and Returns on Capital, Economic Synopses 2015-no. 19, Federal Reserve Bank of St. Louis.
- Gordon, R. (2012), "Is US economic growth over? Faltering innovation confronts the six headwinds", National Bureau of Economic Research Working Paper 18315.
- Greenwald, D.L., Lettau, M. and Ludvigson, S.C. (2020), "How the wealth was won: factors shares as market fundamentals", NBER Working Paper No. 25769.
- Greenwood, R., Hanson, S.G. and Stein, J.C. (2015), "A comparative-advantage approach to government debt maturity", *Journal of Finance*, Vol. 70 No. 4, pp. 1683-1722.
- Gutierrez, G. and Philippon, T. (2017), "Investment-less growth: an empirical investigation", Brookings Papers on Economic Activity, Fall 2017, pp. 89-190.
- Hall, R. (2017a), "Low interest rates: causes and consequences", International Journal of Central Banking, Vol. 13 No. 3, pp. 103-117.
- Hall, R. (2017b), "The role of the growth of risk-averse wealth in the decline of the safe real interest rate", Stanford University Working Paper.
- Jiang, Z., Lustig, H., Van Nieuwerburgh, S. and Xiaolan, M.Z. (2020), "The US public debt valuation puzzle", Northwestern University Working Paper.
- Karabarbounis, L. and Neiman, B. (2014), "The global decline of the labor share", Quarterly Journal of Economics, Vol. 129 No. 1, pp. 61-103.

Kaymak, B. and Poschke, M. (2016),	, "The evolution of wea	lth inequality over	half a century: the role of
taxes, transfers and technolo	ogy", Journal of Monet	ary Economics, Vo	ol. 77 No. C, pp. 1-25.

- Koopmans, T.C. (1965), "On the concept of optimal economic growth", The Econometric Approach to Development Planning Chap. 4, North-Holland Publishing, Amsterdam, pp. 225-87.
- Kozlowski, J., Veldkamp, L. and Venkateswaran, V. (2018), "The tail that keeps the riskless rate low", National Bureau of Economic Research Macroeconomics Annual, Vol. 33, pp. 253-283.
- Kozlowski, J., Veldkamp, L. and Venkateswaran, V. (2020), "The tail that wags the economy: beliefs and persistent stagnation", *Journal of Political Economy*, Vol. 128 No. 8, pp. 2839-2879.
- Krishnamurthy, A. and Vissing-Jorgensen, A. (2011), "The effects of quantitative easing on interest rates: channels and implications for policy", *Brookings Papers on Economic Activity*, Fall 2011, pp. 215-65.
- Krishnamurthy, A. and Vissing-Jorgensen, A. (2012), "The aggregate demand for Treasury debt", Journal of Political Economy, Vol. 120, pp. 233-267.
- Kuttner, K.N. (2018), "Outside the box: unconventional monetary policy in the Great recession and beyond", *Journal of Economic Perspectives*, Vol. 32 No. 4, pp. 121-146.
- Lisack, N., Rana, S. and Thwaites, G. (2017), "Demographic trends and the real interest rate", Bank of England Staff Working Paper No. 701.
- Liu, E., Mian, A. and Amir, S. (2019), "Low interest rates, market power, and productivity growth", Princeton University Working Paper.
- Lukas, R. (1978), "Asset price in an exchange economy", Econometrica, Vol. 46, pp. 1429-1445.
- Lunsford, K. and West, K.D. (2019), "Some evidence on secular drivers of US safe real rates", American Economic Journal: Macroeconomics, Vol. 11 No. 4, pp. 113-139.
- Ma, L., Mello, A.S. and Wu, Y. (2020), "First-mover advantage, time to finance, and cash holdings", *Journal of Corporate Finance*, Vol. 62, 101584.
- Mehra, R. (2003), "The equity premium: why is it a puzzle?", *Financial Analysts Journal*, Vol. 59 No. 1, pp. 54-69.
- Mehra, R. and Prescott, E.C. (1985), "The equity premium: a puzzle", Journal of Monetary Economics, Vol. 15 No. 2, pp. 145-161.
- Nagel, S. (2016), "The liquidity premium of near-money assets", *Quarterly Journal of Economics*, Vol. 131, pp. 1927-1971.
- Powell, J.H. (2020), "New economic challenges and the fed's monetary policy review, a virtual speech at the annual jackson hole economic symposium sponsored by the federal Reserve Bank of Kansas city" available at: https://www.federalreserve.gov/newsevents/speech/powell20200827a.htm.
- Rachel, L. and Smith, T.D. (2017), "Are low real interest rates here to stay?", International Journal of Central Banking, Vol. 13 No. 3, pp. 1-42.
- Ramsey, F.P. (1928), "A mathematical theory of saving", Economic Journal, Vol. 38 No. 152, pp. 543-559.
- Rietz, T.A. (1988), "The equity risk premium: a solution", Journal of Monetary Economics, Vol. 22 No. 1, pp. 117-131.
- Rogoff, K. (2017), "Dealing with monetary paralysis at the zero bound", Journal of Economic Perspectives, Vol. 31 No. 3, pp. 47-66.
- Rubinstein, M. (1976), "The valuation of uncertain income streams and the pricing of options", Bell Journal of Economics, Vol. 7 No. 2, pp. 407-425.
- Sajedi, R. and Thwaites, G. (2016), "Why are real interest rates so low? The role of the relative price of investment goods", *IMF Economic Review*, Vol. 64 No. No. 4, pp. 635-659.
- Samuelson, P.A. (1958), "An exact consumption-loan model of interest with or without the social contrivance of money", *Journal of Political Economy*, Vol. 66 No. 6, pp. 467-82.

- Schmitt-Grohe, S. and Uribe, M., "Liquidity traps: an interest-rate-based exit strategy", Manchester School, Vol. 82 S1, pp. 1-14.
- Solow, R.M. (1956), "A contribution to the theory of economic growth", Quarterly Journal of Economics, Vol. 70 No. 1, pp. 65-94.
- Summers, L. (2014), "US economic prospects: secular stagnation, hysteresis, and the zero lower bound", *Business Economics*, Vol. 49 No. 2, pp. 65-73.
- Summers, L. (2018), "Why the fed needs a new monetary framework", The Hutchins Center on Fiscal and Monetary Policy at Brookings Report: Rethinking the Fed's 2 Percent Inflation Target.
- Taylor, J. (1993), "Discretion versus policy rules in practice", Carnegie-Rochester Conference Series on Public Policy, Vol. 39, pp. 195-214.
- Uribe, M. (2019), "The Neo-fisher effect: econometric evidence from empirical and optimizing models", Columbia University Working Paper.
- Van Binsbergen, J.H., van William, D. and Grotteria, M. (2019), "Risk free interest rates", NBER Working Paper 26138.
- Wang, P., Wen, Y. and Xu, Z. (2017), "Two-way capital flows and global imbalances", *Economic Journal*, Vol. 127, pp. 229-269.
- Wang, Y., Whited, T.M., Wu, Y. and Xiao, K. (2019), "Bank market power and monetary policy transmission: evidence from a structural estimation", University of Michigan Working Paper.
- Weil, P. (1989), "The equity premium puzzle and the risk-free rate puzzle", Journal of Monetary Economics, Vol. 24 No. 3, pp. 401-421.
- Williams, J.C. (2014), "Monetary policy at the zero lower bound: putting theory into practice", Paper Presented at the Hutchins Center on Fiscal and Monetary Policy at Brookings.
- Williams, J.C. (2016), Monetary Policy in a Low R-Star World, Federal Reserve Bank of San Francisco Economic Letter 2016-23.
- Williamson, S. (2016), Neo-Fisherism: A Radical Idea, or the Most Obvious Solution to the Low-Inflation Problem? Regional Economist July 2016, Federal Reserve Bank at St. Louis.
- Williamson, S. (2018), "Inflation control: do central bankers have it right?", Federal Reserve Bank of St. Louis Review, Vol. 100 No. 2, pp. 127-50.
- Williamson, S. (2019), "Neo-fisherism and inflation control", *Canadian Journal of Economics*, Vol. 52, pp. 882-913.
- Yi, K.M. and Zhang, J. (2017), "Understanding global trends in long-run real interest rates", *Economic Perspectives*, Vol. 41 No. 2, pp. 1-20.

# **Corresponding author**

Youchang Wu can be contacted at: ywu2@uoregon.edu

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com